Author’s response to reviews

Title: Non-invasive evaluation of myocardial reperfusion by transthoracic Doppler echocardiography and single-photon emission computed tomography in patients with anterior acute myocardial infarction

Authors:

Egle Sadauskiene (egle.sadauskiene@santa.lt)
Diana Zakarkaite (diana.zakarkaite@santa.lt)
Ligita Ryliskyte (ligita.ryliskyte@santa.lt)
Jelena Čelutkiene (jelena.celutkiene@santa.lt)
Alfredas Rudys (alfredas.rudys@santa.lt)
Sigita Aidietiene (sigita.aidietiene@santa.lt)
Aleksandras Laucevicius (alius@cardio.lt)

Version: 3 Date: 22 May 2011

Author’s response to reviews:

Vilnius, 22nd May, 2011

Dear Independent Editorial Production Team,

Thank You for the sent request regarding formatting changes.

We made major revisions as well as minor revisions:

1. The list of the authors in the manuscript was updated exactly as they are in the submission system, both in style and order.

2. Abstract was made identical in the manuscript and the submission system.

3. Authors’ contribution was updated, also we indicated that all authors read and approved the final manuscript.

4. We formatted manuscript references by eliminating superscript numbers.

5. We removed shading, colours within tables, and the visible vertical lines from the tables. We highlighted the previously shaded text by making it bold.

6. We removed all highlighting and tracking from the manuscript.

7. Abbreviations were formatted into a paragraph, now they occur in alphabetical order.

8. References now do not contain the term ‘et al.’, if there are fewer than 30 authors.
9. Figures are corrected: the image files do not include the title or figure number.

10. We checked typography as much as we could. In case typographical errors were missed we kindly ask to correct them

Best regards

On behalf of the authors

Egle Sadauskiene